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23 September 2012
MEMORANDUM FOR CENWP-OP, ATTN: Operations Biologist
SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 38 of 2012, which covers the period from 16 through 22 September 2012. 
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 104.3 to 134.8 kcfs.  Daily average project tailwater ranged from 9.9’ to 12.4’ msl.  Daily average powerhouse forebay elevation ranged from 74.0’ to 75.3’ msl.  Daily average water temperature ranged from 67ºF to 68ºF. Secchi disk measurements remained at 7.0’.
1.2 Spill Bays 1 and 18, which provide attraction flow for Cascades Island and B-branch, respectively, are open 6” during daylight hours. Daily average spill ranged from 1.2 to 1.3 kcfs.
1.3 Unit Operation.  Powerhouse One is the priority powerhouse as of 16 September. 
1.3.1 Unit 15 went out of service at 0006 on 27 August for T12 bi-annual.

1.3.2 Unit 16 went out of service at 0006 on 27 August for T12 bi-annual.
1.3.3 Unit 17 went out of service at 0735 on 01 August for air gap monitor installation.
1.3.4 Unit 18 went out of service at 0006 on 27 August for T12 and annual maintenance.
1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore (WS) fish ladder.  The units may be placed on standby, as necessary, during nighttime hours to float debris. 
1.5 Gatewell drawdowns are measured on Tuesdays and more frequently as needed.
2 ACTIVITIES
2.1 Auxiliary Water System Closures.  No AWS closures occurred this reporting week.
2.2 STS/VBS inspections.  No STS/VBS inspection occurred this reporting week. 
2.2.1 Table 1.  STS and VBS inspection.  

	Unit
	Date
	Run Hours
	Comments

	11,12,18
	N/A
	N/A
	STSs installed

	13-17
	N/A
	N/A
	STSs removed


2.3 Dewatering and Fish Salvage. The following dewatering or fish salvage activities occurred this week.
2.3.1 On 20 September Project Biologists recovered 1 smallmouth bass, 26 sculpin, and 1 lamprey from the Unit 17 tail logs. All fishes were released into the tailrace in good condition.

3 RESEARCH
3.1 Battelle/PNNL – Survival Study. Hydrophones remain installed.
3.2 CRITFC – AFF.  Annual Chinook, Coho, Sockeye and Steelhead sampling in the AFF continues. 
3.3 NOAA – Lamprey. LPS count network installed. Trap and haul from the CI LPS to the Stevenson release site concluded on 20 September.
3.4 PSMFC – Smolt Monitoring Program.  Juvenile fish condition and index monitoring continues. Truck releases of summer out-migrants originating from McNary, Lower Granite, Little Goose, and Lower Monumental dams began on 18 August. Fallbacks were recorded crossing the separator bars this week: 3 salmon and 1 unknown salmonid. Fallback observations are available at the FPC website: http://www.fpc.org/adultsalmon/bonkeltcounts.htm.
3.5 WDFW – AFF. Sampling for alternative gear study control group continues.
3.6 WDFW – Fish Counting.  Visual counting continues. Nighttime video lamprey counting continues.
3.7 WDFW – Tule Counting. Visual counting continues.
4 FISHWAYS
4.1 Project biologists inspected 16 through 22 September. 
4.2 Fish Passage Plan Observations.
4.2.1 Table 2.  FPP Out of Criteria items.
	Date
	Location
	FPP Items
	Cause
	Response

	BRADFORD ISLAND (BI)

	9/22
	A-branch
	Ent. differential < 1.0
	Unknown
	None

	9/20-21
	B-branch
	Ent. differential > 2.0
	FV 4-4 damaged
	Repairs scheduled

	CASCADES ISLAND (CI) – Nothing to report.

	WASHINGTON SHORE (WS) 

	9/21-22
	Weir 37
	Water level < 0.9’
	Unknown
	None


4.2.2 PH1CC diffuser FG2-19 is partially open.  Repairs will occur when the PH1CC is dewatered. The next possible time period will be 2013/14 winter maintenance.
4.2.3 A new mechanical velocity meter has been fabricated for PH2 and is scheduled for installation.
4.3 Fishway Temperatures.  Temperatures are taken 01 March through 30 November. BON site specific temperatures are unavailable this week due to technical difficulties. The temperatures below are provided via this website and show the averages for BON forebay.
	Date
	Temperature ((F)

	09/16
	65.9

	09/17
	65.8

	09/18
	66.4

	09/19
	66.9

	09/20
	66.9

	09/21
	66.9

	09/22
	67.1


4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  
4.5 Adult Fishways. 
4.5.1 Calibration was checked on 14 September.
4.5.2 10 of 12 floating orifice gates (FOGs) are installed. Two gates remain out for maintenance. 
4.5.3 SLEDs are removed from all entrances.
4.5.4 All ladders remain in service.
4.5.5 On 09 August, FV 4-4 was discovered to be broken in the full open position. Upon ROV inspection, the gate appears to be sheared from its attachment point on the wall. The ladder is currently being operated as close to FPP criteria as possible. Repairs are scheduled to start 01 October.
4.5.6 The AFF remains in service.
4.6 Juvenile Fishways.
4.6.1 DSM2 remains in service.
4.6.2 The PH1 ice and trash sluiceway (ITS) remains in service. Manual chaingates 1A and 1B remain open and are both set to 70’ msl. Automated chaingates 3B, 6C and 10B are open and functioning in auto.
4.6.3 The hydrocannon remains in service.
4.6.4 The B2CC closed at 0009 on 16 September. 
4.6.5 The B2CC avian array was removed to facilitate barge access for B-branch emergency repairs. The array will be reinstalled during winter of 2013.
4.7 Lamprey Fishways.
4.7.1 The Cascades Is. entrance weir remains installed.
4.7.2 The Cascades Is. LPS was taken out of service on 20 September. Bradford Island and WA Shore LPSs remain in service. 
5 CONSTRUCTION
5.1 Seawater Recirculation System Removal (JMF). Removal continues.
5.2 Cascades Island LPS Continuation. Construction continues. This expansion will remove the evaluation trap box and continue the LPS flume to a volitional exit near the CI fishway exit.
5.3 Gantry 7 Rehabilitation. The Powerhouse 2 intake gantry crane went out of service on 01 August for rehabilitation. It is scheduled to return to service in February 2013.
5.4 Spillbay 17. Final repairs to the hoist and the installation of new wire rope began on 17 September.
6 HAZMAT, SPILLS AND CLEANUP. Nothing to report.
7 TELETYPES CURRENTLY IN EFFECT
7.1 A teletype sent on 23 August updates the operation of BON project PH2 unscreened units (Units 13-17) to pass flow. This operation is necessary due to the rehabilitation of Gantry 7. During the Gantry 7 outage, the project will not be able to remove or install screens. Additionally, limit flow through PH2 screened units (Units 11, 12, and 18) to a maximum of 15 kcfs per unit. The intent of this limitation is to minimize the change of damaging an operating screen during the Gantry 7 outage, which would force the unit out of service until February 2013.
7.2 A teletype sent on 23 August states that effective 0001 hours on September 1, the PH2 Corner Collector (B2CC) will be operated any time one or more unscreened units are operating at PH2, with the exception of September 4 (0700-1700), when it will be closed for repairs. The B2CC will be closed on September 16 through November 11 for repairs. Additionally, on 16 September PH1 will become the priority powerhouse.  In November, MU18 will be taken out of service and the VBS will be removed from MU11 and MU12, when the downstream migrant channel (DSM2) is dewatered for installation of a lamprey passage system. At that time, those units will be operated last in the priority order. 

7.3 A teletype sent on 17 September updates the spill priority list order for non-fish related spill season for BON with associated total dissolved gas spill caps.

7.4 A teletype sent on 17 September requests that the BON pool be operated within a 1.5’ band, as a soft constraint from 0600 to 1800 on 18 September and as a hard constraint from 1801 on 18 September until 1800 on 21 September in order to limit pool fluctuation for treaty fishing. 
JERRY CARROLL
Operations Project Manager
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